Author's response to reviews

Title: Frontalis Suspension Surgery to Treat Patients with Essential Blepharospasm and Apraxia of Eyelid Opening - Technique and Results -

Authors:

Chrisanthis Karapantzou (chrisanthis24@yahoo.gr)
Dirk Dressler (Dressler.Dirk@mh-hannover.de)
Saskia Rohrbach (saskia.rohrbach@googlemail.com)
Rainer Laskawi (rlaskaw@gwdg.de)

Version: 3
Date: 29 September 2014

Author's response to reviews:

Dear Editors,

Thank you for sending the reviewers comments.
We integrated the remarks of the reviewers (Ref. 1, Ref. 2 no changes needed) in the manuscript.
The additional new text is printed in blue (see manuscript text).
The following points are modified:
1. Limitations: the remarks of Ref 1. are integrated and mentioned in the discussion as suggested (see marked blue text)
2. The manuscript has been re-read and is now again corrected by a native English speaker.
3. On page 6, a small correction has been carried out. The calculated numbers are now presented with an accuracy of one digit after the decimal point (blue marked numbers)

Thank you for the possibility to publish our work in H&F Medicine!

Yours sincerely

Prof. Rainer Laskawi
Corresponding author